Author’s response to reviews

Title: Physicians’ choices of dual practices and the effects on labor supply in public hospitals: Results from a register based study.

Authors:

Karl Arne Johannessen Dr. PhD (karlarne.johannessen@online.no)
Terje P Hagen Professor (t.p.hagen@medisin.uio.no)

Version: 2 Date: 29 October 2013

Author’s response to reviews: see over
Dear Flory Mae Calumpita, Journal Editorial Office, Biomed Central

Thank you so much for your email of 24th of October 2013.
I regret that I was not able to respond immediately, but I have been on a congress in Qatar the last week. We sincerely regret that we had misinterpreted the instructions and uploaded the figures in the main file.

We have done the following changes:

- Included a title page. This now includes the email address of both authors, and we have added Norway as country for both.
- We have added an Ethics paragraph after the Author contributions stating:
  
  The study was subject to notification according to the Norwegian Personal Data Act. Notification was submitted to Norwegian Social Science Data Services (NSD) 5th of May 2008. Project number: 19192.

  This fully covers the legal and ethical regulations in this type of studies under Norwegian law.

- We have deleted the figures from the main document and upload them as six separate files. We have included Legends in the paper.
- We have no Acknowledgements for this paper.
- We have revised Table 1 and deleted the vertical lines and merged cells.

We hope that this makes this version is according to the journal specifications.

Best Regards

Karl-Arne Johannessen
MD, PhD